Author's response to reviews

Title: Effectiveness of polaprezinc for low-dose aspirin-induced small-bowel mucosal injuries as evaluated by capsule endoscopy: a pilot randomized controlled study

Authors:

Ikue Watari (ikuen@hiroshima-u.ac.jp)
Shiro Oka (oka4683@hiroshima-u.ac.jp)
Shinji Tanaka (colon@hiroshima-u.ac.jp)
Taiki Aoyama (aochin-blue@hiroshima-u.ac.jp)
Hiroki Imagawa (h-imagawa522@hiroshima-u.ac.jp)
Takayoshi Shishido (sisido@hiroshima-u.ac.jp)
Shigeto Yoshida (yoshida7@hiroshima-u.ac.jp)
Kazuaki Chayama (chayama@mba.ocn.ne.jp)

Version: 5 Date: 26 March 2013

Author's response to reviews:

March 26, 2013

Dear Dr. Foote:

We thank you for the thorough review of our manuscript, “Effectiveness of polaprezinc for low-dose aspirin-induced small-bowel mucosal injuries as evaluated by capsule endoscopy: a pilot randomized controlled study,” which we submitted for publication as an Original Article (MS: 7491729588479121). We have taken all of the reviewers’ comments into account and have made adjustments to the text accordingly. We herein submit our revised manuscript. As you will see, the revisions are marked by single underlining.

We feel that our revisions are suitable in light of the reviewers’ comments, and we believe the revised paper is significantly improved over the paper we submitted initially. We hope that the revised version is now acceptable for publication in BMC Gastroenterology.

Please do not hesitate to let us know whether there are any further changes we should make to the manuscript.

Yours sincerely,

Ikue Watari, MD
e-mail: ikuen@hiroshima-u.ac.jp

All related correspondence should be sent to: Shiro Oka, MD, PhD.
Department of Endoscopy, Hiroshima University Hospital, 1-2-3 Kasumi, Minami-Ku, Hiroshima 734-8551, Japan
Telephone: 81-82-257-5193, Facsimile: 81-82-257-5193
e-mail: oka4683@hiroshima-u.ac.jp